\documentclass[kind=GCE,a4j,bachelor,landscape,viewer,color,check]{edb-syl}
\def\�N�x{2011}
\def\�a�N�x{23}
\begin{document}

\Englishfalse
%%%% 0:221176
\begin{edb:lecture}%
\EdbEID{221176}\EdbEOID{552074}%
\���false%
\����ID{218270:221176}
\���ƊT�v���{166225}%
\����{�p��}{English}%
\���{���ʉp��}{Thematic English}%
\�Ώ�{�H}{(��B)1�N}%
\�J�u{��}{��}{5�E6}%
\�S������{\������������{}{���΍u�t}{\EdbSMG{�p�g���b�N}{}{�W�F�t}}}%
\EdbColDef{edb:lecture.goal}{1}{E}{To gain a broader understanding of current events and the language necessary to discuss it}%
\EdbColDef{edb:lecture.goal}{1}{J}{To gain a broader understanding of current events and the language necessary to discuss it}%
\EdbColDef{edb:lecture.plan}{1}{E}{The first weekly topic sheet will be handed out \& groups formed}%
\EdbColDef{edb:lecture.plan}{1}{J}{The first weekly topic sheet will be handed out \& groups formed}%
\EdbColDef{edb:lecture.plan}{2}{E}{The first topic will be chosen.\ The second weekly topic sheet will be handed out}%
\EdbColDef{edb:lecture.plan}{2}{J}{The first topic will be chosen.\ The second weekly topic sheet will be handed out}%
\EdbColDef{edb:lecture.plan}{3}{E}{Grammar and expressions in the first topic will be practiced.\ The second topic will be decided.}%
\EdbColDef{edb:lecture.plan}{3}{J}{Grammar and expressions in the first topic will be practiced.\ The second topic will be decided.}%
\EdbColDef{edb:lecture.plan}{4}{E}{Issues and themes addressed in the first topic will be discussed.\ Third weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{4}{J}{Issues and themes addressed in the first topic will be discussed.\ Third weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{5}{E}{Grammar and expressions in the second topic will be practiced.\ The third topic will be decided.}%
\EdbColDef{edb:lecture.plan}{5}{J}{Grammar and expressions in the second topic will be practiced.\ The third topic will be decided.}%
\EdbColDef{edb:lecture.plan}{6}{E}{Issues and themes addressed in the second topic will be discussed.\ Fourth weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{6}{J}{Issues and themes addressed in the second topic will be discussed.\ Fourth weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{7}{E}{Grammar and expressions in the third topic will be practiced.\ The fourth topic will be decided.}%
\EdbColDef{edb:lecture.plan}{7}{J}{Grammar and expressions in the third topic will be practiced.\ The fourth topic will be decided.}%
\EdbColDef{edb:lecture.plan}{8}{E}{Issues and themes addressed in the third topic will be discussed.\ The fifth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{8}{J}{Issues and themes addressed in the third topic will be discussed.\ The fifth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{9}{E}{\EdbAlert{���Ԏ���}}%
\EdbColDef{edb:lecture.plan}{9}{J}{���Ԏ���}%
\EdbColDef{edb:lecture.plan}{10}{E}{Grammar and expressions used in the fourth topic will be practiced.\ The fifth topic will be chosen.}%
\EdbColDef{edb:lecture.plan}{10}{J}{Grammar and expressions used in the fourth topic will be practiced.\ The fifth topic will be chosen.}%
\EdbColDef{edb:lecture.plan}{11}{E}{Issues and themes addressed in the fourth topic will be discussed.\ The sixth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{11}{J}{Issues and themes addressed in the fourth topic will be discussed.\ The sixth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{12}{E}{Grammar and expressions used in the fifth topic will practiced.\ The sixth topic will be chose.}%
\EdbColDef{edb:lecture.plan}{12}{J}{Grammar and expressions used in the fifth topic will practiced.\ The sixth topic will be chose.}%
\EdbColDef{edb:lecture.plan}{13}{E}{Issues and themes addressed in the fifth topic will be discussed.}%
\EdbColDef{edb:lecture.plan}{13}{J}{Issues and themes addressed in the fifth topic will be discussed.}%
\EdbColDef{edb:lecture.plan}{14}{E}{Grammar and expressions used in the sixth topic will be practiced.}%
\EdbColDef{edb:lecture.plan}{14}{J}{Grammar and expressions used in the sixth topic will be practiced.}%
\EdbColDef{edb:lecture.plan}{15}{E}{\EdbAlert{��������}}%
\EdbColDef{edb:lecture.plan}{15}{J}{��������}%
\EdbColDef{edb:lecture.plan}{16}{E}{\EdbAlert{���K;DVD}}%
\EdbColDef{edb:lecture.plan}{16}{J}{���K;DVD}%
\EdbColDef{edb:lecture.style}{1}{@}{101768}%
\EdbColDef{edb:lecture.style}{1}{E}{�����p��}%
\EdbColDef{edb:lecture.style}{1}{J}{�����p��}%
\EdbColDef{edb:lecture.unit}{1}{E}{1}%
\EdbColDef{edb:lecture.unit}{1}{J}{1}%
\EdbColDef{edb:lecture.target}{1}{E}{Students will be made aware of global current events and trends and will be exposed to language and expressions that will help them discuss the issues.}%
\EdbColDef{edb:lecture.target}{1}{J}{Students will be made aware of global current events and trends and will be exposed to language and expressions that will help them discuss the issues.}%
\EdbColDef{edb:lecture.outline}{1}{E}{Every week a different topic is discussed.\ Those topics will be chosen by the students.}%
\EdbColDef{edb:lecture.outline}{1}{J}{Every week a different topic is discussed.\ Those topics will be chosen by the students.}%
\EdbColDef{edb:lecture.notice}{1}{E}{Bring a good dictionary.\ We will be using natural college level English.}%
\EdbColDef{edb:lecture.notice}{1}{J}{Bring a good dictionary.\ We will be using natural college level English.}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{\EdbAlert{�앶$\cdot{}$�ԓx$\cdot{}$��������}}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{�앶$\cdot{}$�ԓx$\cdot{}$��������}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{\EdbAlert{�L}}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{�L}%
\EdbColDef{edb:lecture.textbook}{1}{E}{\EdbAlert{�����D���Ǝ��Ƀv�����g��z�z���܂��D}}%
\EdbColDef{edb:lecture.textbook}{1}{J}{�����D���Ǝ��Ƀv�����g��z�z���܂��D}%
\EdbColDef{edb:lecture.note}{1}{E}{\EdbAlert{���J���L�������ł͎��ƉȖ�:{\EdbZSp}�p��(2)}}%
\EdbColDef{edb:lecture.note}{1}{J}{���J���L�������ł͎��ƉȖ�:{\EdbZSp}�p��(2)}%
\EdbColDef{edb:lecture.contact}{1}{J}{\�A����l{}{�p�g���b�N . }{}}%
\��������{�p��(2)}%
\������{���ʉp��}%
\end{edb:lecture}%

\Englishtrue
%%%% 0:221176
\begin{edb:lecture}%
\EdbEID{221176}\EdbEOID{552074}%
\���false%
\����ID{218270:221176}
\���ƊT�v���{166225}%
\����{English}{�p��}%
\���{Thematic English}{���ʉp��}%
\�Ώ�{�H}{(��B)1�N}%
\�J�u{��}{��}{5�E6}%
\�S������{\������������{}{Part-time Lecturer}{\EdbGMS{JEFFERY}{}{PATRICK}}}%
\EdbColDef{edb:lecture.goal}{1}{J}{To gain a broader understanding of current events and the language necessary to discuss it}%
\EdbColDef{edb:lecture.goal}{1}{E}{To gain a broader understanding of current events and the language necessary to discuss it}%
\EdbColDef{edb:lecture.plan}{1}{J}{The first weekly topic sheet will be handed out \& groups formed}%
\EdbColDef{edb:lecture.plan}{1}{E}{The first weekly topic sheet will be handed out \& groups formed}%
\EdbColDef{edb:lecture.plan}{2}{J}{The first topic will be chosen.\ The second weekly topic sheet will be handed out}%
\EdbColDef{edb:lecture.plan}{2}{E}{The first topic will be chosen.\ The second weekly topic sheet will be handed out}%
\EdbColDef{edb:lecture.plan}{3}{J}{Grammar and expressions in the first topic will be practiced.\ The second topic will be decided.}%
\EdbColDef{edb:lecture.plan}{3}{E}{Grammar and expressions in the first topic will be practiced.\ The second topic will be decided.}%
\EdbColDef{edb:lecture.plan}{4}{J}{Issues and themes addressed in the first topic will be discussed.\ Third weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{4}{E}{Issues and themes addressed in the first topic will be discussed.\ Third weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{5}{J}{Grammar and expressions in the second topic will be practiced.\ The third topic will be decided.}%
\EdbColDef{edb:lecture.plan}{5}{E}{Grammar and expressions in the second topic will be practiced.\ The third topic will be decided.}%
\EdbColDef{edb:lecture.plan}{6}{J}{Issues and themes addressed in the second topic will be discussed.\ Fourth weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{6}{E}{Issues and themes addressed in the second topic will be discussed.\ Fourth weekly topic sheets will be handed out.}%
\EdbColDef{edb:lecture.plan}{7}{J}{Grammar and expressions in the third topic will be practiced.\ The fourth topic will be decided.}%
\EdbColDef{edb:lecture.plan}{7}{E}{Grammar and expressions in the third topic will be practiced.\ The fourth topic will be decided.}%
\EdbColDef{edb:lecture.plan}{8}{J}{Issues and themes addressed in the third topic will be discussed.\ The fifth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{8}{E}{Issues and themes addressed in the third topic will be discussed.\ The fifth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{9}{J}{���Ԏ���}%
\EdbColDef{edb:lecture.plan}{9}{E}{\EdbAlert{���Ԏ���}}%
\EdbColDef{edb:lecture.plan}{10}{J}{Grammar and expressions used in the fourth topic will be practiced.\ The fifth topic will be chosen.}%
\EdbColDef{edb:lecture.plan}{10}{E}{Grammar and expressions used in the fourth topic will be practiced.\ The fifth topic will be chosen.}%
\EdbColDef{edb:lecture.plan}{11}{J}{Issues and themes addressed in the fourth topic will be discussed.\ The sixth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{11}{E}{Issues and themes addressed in the fourth topic will be discussed.\ The sixth topic sheet will be handed out.}%
\EdbColDef{edb:lecture.plan}{12}{J}{Grammar and expressions used in the fifth topic will practiced.\ The sixth topic will be chose.}%
\EdbColDef{edb:lecture.plan}{12}{E}{Grammar and expressions used in the fifth topic will practiced.\ The sixth topic will be chose.}%
\EdbColDef{edb:lecture.plan}{13}{J}{Issues and themes addressed in the fifth topic will be discussed.}%
\EdbColDef{edb:lecture.plan}{13}{E}{Issues and themes addressed in the fifth topic will be discussed.}%
\EdbColDef{edb:lecture.plan}{14}{J}{Grammar and expressions used in the sixth topic will be practiced.}%
\EdbColDef{edb:lecture.plan}{14}{E}{Grammar and expressions used in the sixth topic will be practiced.}%
\EdbColDef{edb:lecture.plan}{15}{J}{��������}%
\EdbColDef{edb:lecture.plan}{15}{E}{\EdbAlert{��������}}%
\EdbColDef{edb:lecture.plan}{16}{J}{���K;DVD}%
\EdbColDef{edb:lecture.plan}{16}{E}{\EdbAlert{���K;DVD}}%
\EdbColDef{edb:lecture.style}{1}{@}{101768}%
\EdbColDef{edb:lecture.style}{1}{J}{�����p��}%
\EdbColDef{edb:lecture.style}{1}{E}{�����p��}%
\EdbColDef{edb:lecture.unit}{1}{J}{1}%
\EdbColDef{edb:lecture.unit}{1}{E}{1}%
\EdbColDef{edb:lecture.target}{1}{J}{Students will be made aware of global current events and trends and will be exposed to language and expressions that will help them discuss the issues.}%
\EdbColDef{edb:lecture.target}{1}{E}{Students will be made aware of global current events and trends and will be exposed to language and expressions that will help them discuss the issues.}%
\EdbColDef{edb:lecture.outline}{1}{J}{Every week a different topic is discussed.\ Those topics will be chosen by the students.}%
\EdbColDef{edb:lecture.outline}{1}{E}{Every week a different topic is discussed.\ Those topics will be chosen by the students.}%
\EdbColDef{edb:lecture.notice}{1}{J}{Bring a good dictionary.\ We will be using natural college level English.}%
\EdbColDef{edb:lecture.notice}{1}{E}{Bring a good dictionary.\ We will be using natural college level English.}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{�앶$\cdot{}$�ԓx$\cdot{}$��������}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{\EdbAlert{�앶$\cdot{}$�ԓx$\cdot{}$��������}}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{�L}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{\EdbAlert{�L}}%
\EdbColDef{edb:lecture.textbook}{1}{J}{�����D���Ǝ��Ƀv�����g��z�z���܂��D}%
\EdbColDef{edb:lecture.textbook}{1}{E}{\EdbAlert{�����D���Ǝ��Ƀv�����g��z�z���܂��D}}%
\EdbColDef{edb:lecture.note}{1}{J}{���J���L�������ł͎��ƉȖ�:{\EdbZSp}�p��(2)}%
\EdbColDef{edb:lecture.note}{1}{E}{\EdbAlert{���J���L�������ł͎��ƉȖ�:{\EdbZSp}�p��(2)}}%
\EdbColDef{edb:lecture.contact}{1}{E}{\�A����l{}{PATRICK . }{}}%
\��������{�p��(2)}%
\������{���ʉp��}%
\end{edb:lecture}%

\end{document}
